

Essau



The Impacts of Online Experience on Health and Well-Being: The Overlooked Aesthetic Dimension

Tim Gorichanaz ¹, Alexandros A. Lavdas ^{2,3,4}, Michael W. Mehaffy ⁵ and Nikos A. Salingaros ^{6,7,*}

- ¹ College of Computing and Informatics, Drexel University, 3141 Chestnut Street, Philadelphia, PA 19104, USA; gorichanaz@drexel.edu
- ² Eurac Research, Institute for Biomedicine, Affiliated Institute of the University of Lübeck, Via Galvani 31, 39100 Bolzano, Italy; alexandros.lavdas@eurac.edu
- ³ The Human Architecture & Planning Institute, Inc., 43 Bradford St., Concord, MA 01742, USA
- ⁴ Department of Psychology, Webster University, Athens Campus, 9 Ipitou Street, 10557 Athens, Greece
- ⁵ Sustasis Foundation, P.O. Box 2579, White Salmon, WA 98672, USA; michael.mehaffy@gmail.com
- ⁶ Departments of Mathematics and Architecture, University of Texas at San Antonio, One UTSA Circle, San Antonio, TX 78249, USA
- ⁷ Galaxylabs, Laurenzenvorstadt 69, 5000 Aarau, Switzerland
- * Correspondence: salingar@gmail.com

Abstract: It is well-recognized that online experience can carry profound impacts on health and well-being, particularly for young people. Research has already documented influences from cyberbullying, heightened feelings of inadequacy, and the relative decline of face-to-face interactions and active lifestyles. Less attention has been given to the health impacts of aesthetic experiences of online users, particularly gamers and other users of immersive virtual reality (VR) technologies. However, a significant body of research has begun to document the surprisingly strong yet previously unrecognized impacts of aesthetic experiences on health and well-being in other arenas of life. Other researchers have used both fixed laboratory and wearable sensors and, to a lesser extent, user surveys to measure indicators of activation level, mood, and stress level, which detect physiological markers for health. In this study, we assessed the evidence that online sensorial experience is no less important than in the physical world, with the capacity for both harmful effects and salutogenic benefits. We explore the implications for online design and propose an outline for further research.

Keywords: virtual world; virtual reality; augmented reality; immersive environment; visual appeal; healing environment; therapeutic environment; beauty

1. Introduction

1.1. From Negative to Positive Impacts on the Health of Online Users

There is currently a growing concern about the potential negative impacts of online experiences on health and well-being, particularly for young people. Among the topics well-studied in the literature are cyberbullying [1], heightened feelings of inadequacy, depression and even suicide [2], and the decline of active lifestyles and healthy face-to-face interactions [3].

There has been less consideration of the potential *positive* impacts of online experience on health and well-being, which is the subject of this paper. In particular, there is emerging evidence of the salutogenic (health-promoting) benefits of certain classes of aesthetic experience, including the experience of structures perceived by the user as beautiful. While the existing literature focuses on naturalistic and urban experiences, there is growing evidence that online experiences can have a similar positive impact.

In this paper, we explore the potential for positive health effects from experiencing aesthetically beautiful virtual worlds. Given that much of the current focus in interface design has been on usability rather than beauty, there is definite room for discussion on



Citation: Gorichanaz, T.; Lavdas, A.A.; Mehaffy, M.W.; Salingaros, N.A. The Impacts of Online Experience on Health and Well-Being: The Overlooked Aesthetic Dimension. *Virtual Worlds* 2023, 2, 243–266. https://doi.org/10.3390/ virtualworlds2030015

Academic Editor: Andreas Riener

Received: 18 July 2023 Revised: 5 August 2023 Accepted: 18 August 2023 Published: 22 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the topic of aesthetics. Even within game design research, there has been little discussion on beauty in game worlds when it might be expected that game worlds that feel beautiful might encourage engagement and a sense of wonder. In addition, there is an important opportunity for aesthetic exploration and refinement of the user's capacities for aesthetic discernment. This contrasts with the potential for passive consumption of crude but exciting aesthetics, resulting in atrophy of the capacity for fine aesthetic discernments. This issue is treated here using a pragmatic approach that draws on results from environmental psychology, biology, and mathematics.

1.2. Shared Aesthetic Judgments and Healing Environments

Furthermore, while there is undoubtedly a socially constructed component of what is perceived to be beautiful, emerging research shows that there are also structural characteristics that are widely shared among different populations (as we will explore in more detail herein). The result is a widely shared set of aesthetic judgments from which a smaller subset of differentiated judgments arise from individual variations and social constructions. Evidence suggests that our widely shared judgments are rooted in our common human evolutionary history, particularly the experience of natural environments, plants, and animals (the so-called "biophilia effect").

The research has documented both positive or "salutogenic" health impacts (from visual experiences judged as beautiful) and negative or "pathogenic" health impacts (from over-exposure to visual experiences judged as ugly and/or stressful). The implication is that designers have a responsibility to provide a basic background structure of salutogenic or health-promoting visual experiences, on which they may choose to create more transitory moments of excitement, stress, or even ugliness.

We, therefore, propose a normative standard for designers to provide a universal visual experience arising from a persistent background of biologically based objective beauty, which continues to be a key ingredient of the natural settings in which our ancestors evolved. It is notable that artificial environments maintained this background effect until the 20th century, after which it was progressively damaged so that it could no longer be relied upon for emotional nourishment. For that reason, it is necessary to re-assess and re-establish this normative standard.

Inherited misunderstandings complicate the creation of genuinely healing environments in the current physical world through commercial processes; a far easier option is to design virtual worlds that embody healing properties, and perhaps this sensibility can later carry over into the physical world [4]. Having positive experiences in virtual worlds triggers salutogenic emotions and unconscious bodily responses in all people, whether they do or do not suffer from any psychological issues [5]. To gain a health benefit, the design of the virtual world should mimic the natural world in a way that induces the biophilic healing effect that comes from plants and special geometries [6–12].

All of these findings point to new opportunities for designers of online experiences and virtual worlds and new ways to meet the needs of online users (including children and other vulnerable populations). The findings also point to the harm that can be done to users and the professional responsibility that online designers, like all designers, owe to their users.

A well-designed virtual experience provides a healthy escape for persons who live in industrial/modernist cities that, as they are built according to currently accepted standards and typologies, have negative effects on their occupants [13]. Those effects are unconscious and lead to anxiety that accumulates over time [14]. Since our society does not usually discuss such low-level persistent mechanisms, they are not consciously perceived by those who are affected. The only clue is a chronic feeling of discontent and general malaise. Periodically, resentment coming from the geometry of the built environment that a sector of society inhabits because of socio-economic constraints can explode into social discontent [15,16].

This is a basic question of public health, not aesthetics per se. Multiple factors affect human health, yet many of them are not sufficiently recognized. Separate from obvious biological and chemical factors, the physical/visual complexity of our environments plays a much larger role in human health and well-being than is usually acknowledged. An environment that the human body perceives as being "alive" in some direct (but hard to pin down) sense helps to trigger positive-valence physio-psychological responses. Over time, this response influences the users' state of health positively (otherwise, environment-induced anxiety generates pathologies) [17].

1.3. Christopher Alexander's "Living Structure" Versus Copying Images

In this discussion, we refer to the work of the architectural theorist Christopher Alexander, whose description of the preferred geometry of place as a "living structure" is not a single and unique state but a complex, multi-scale, multi-category phenomenon (and not only a visual one) [18]. It is important to avoid the reductionist trap of a "silver bullet" or a single solution to the problem of design, which in any case, is always bounded by our own rationality and the limits of model-making. (We could link this point mathematically to a fundamental "incompleteness", as Gödel and others defined it.) To create a "living structure", we need to go through iterations and feedback and gradually evolve healthier and more whole environments.

In previous generations, technological society made the mistake of taking a "silver bullet" approach with narrow criteria, e.g., sanitation, thermal comfort, or access to fresh air and light. These factors are important, but there are many other levels, including the capacity of the distributed agents to make their own evolutionary adaptations. These solutions are contingent upon the capacity of designers and planners to empower them to do so, something woefully neglected in a reductionistic approach. That is what occurred for much of the 20th century, and those standardized design typologies dominate the present day's buildings and cities. As discussed later in this essay, the generative and adaptive nature of Alexander's work versus the industrial/modernist framework emphasizes this point very well.

If we look at, say, the 1926 Dessau Bauhaus and 1927 Stuttgart *Weissenhofsiedlung* building prototypes and the 1933 Athens Charter, they are all about these narrow criteria to be copied superficially as images [19]. Those images—highly "innovative" at the time—were introduced as promises of a healthier and more socially just future. When we built places like this in the past, they failed miserably (e.g., the US "projects", the UK "tower blocks", the Paris "banlieues", etc.). Those environments created according to a reductionistic set of design rules failed to provide for the "organized complexity" and the adaptive capacities of people, the opposite of working with the evolutionary patterns of history and nature.

1.4. AI and Beauty That Evolves through Selection Criteria

Adaptive design is not an act of imposition—despite architectural academia teaching it as such for decades—it is a process, an evolution subject to selection criteria of human adaptation and use. Missing that essential point turns design into a dangerous abstraction. You cannot just apply one narrow mathematical attribute of design and think that it will deal with everything. While true, as far as it goes, this fundamentalism overlooks that there are important and overlooked attributes, without which we are really pursuing a dead end when it comes to healthy human environments!

Given recent developments in artificial intelligence (AI), the question arises of whether virtual worlds generated by AI may spell the end of humans exploring beauty, how to create it, and how to use it to nourish us. Exploring, creating, and using beauty only come about when people discover for themselves how to be human. Many AI implementations at present (e.g., social media news feeds, algorithmically served ads, generative AI content) effectively outsource beauty to machines, and the algorithms push their own selected feed onto children and adolescents, including aesthetic content. This process disempowers individuals who cannot exercise their own natural judgment or develop that capacity in a healthy way. Human beings are Homo faber—"man the maker"—and the present state of affairs discourages and disables people from making. The corporate provider of aesthetic experiences (meant to manipulate but not to nourish) profits by creating unhappiness and even distress in the users. But this situation

is not new to the 21st century; it has been underway since architects in the early 20th century embraced the industrial-modernist aesthetic.

2. Methods and Analysis—Outline of This Essay

The present study collects and relates results from disparate disciplines into a framework for improving the salutogenic experience in virtual worlds. We document and highlight relevant information in the following sequence.

Section 3 summarizes what is known about aesthetics and health outside of the online world, in the discipline of biophilia, etc. This material provides the evidential basis for assessing healing online environments. Section 4 lists in detail what is known about health impacts in the online world. Beginning with the influence of games that provide a visceral experience (Sections 4.1 and 4.2), we introduce the notion of "living structure", following Christopher Alexander (Section 4.3). This section discusses "design patterns", also introduced by Alexander, as documented reusable solutions for achieving adaptive design. It is this "living structure" that is largely responsible for salutogenic effects in both physical and virtual environments. Various design trends have eliminated living structures from the environment, for example, the monotonous repetition that is falsely assumed to be an indispensable feature of industrial production (Section 4.4). However, uniformization and lack of variety have negative consequences, and in some cases, the induced visual stress leads to headaches. Section 4.5 establishes the connection to aesthetics online and in VR worlds. But the point we underline is the absence of beauty in digital design, which could be attributed to several reasons: the lack of teaching resources on how to generate beauty and the confusion that equates an abstract conception of beauty with mechanical efficiency.

Objective beauty in physical architecture is due to specific aesthetic properties and their importance (symmetry, etc.). Design rules for generating objective beauty rely heavily on Christopher Alexander's work on aesthetics and environmental structure (Section 4.6). Designers of virtual worlds naturally copy both design methodology and specific designs from the real world, but this can have opposite effects depending upon whether it satisfies or violates criteria set by Alexander. This section discusses the Roku City screensaver as an example of a viscerally attractive virtual world and explains the design features responsible for its appeal.

Section 4.7 mentions how the new Apple Vision Pro headset combines mixed reality in a way that seems to conform intuitively to our own discussion. One of our principal aims is to introduce virtual worlds as therapeutic settings, a project that has been initiated by various groups. The ability to use real-time feedback from physiological monitors to evolve the appearance of a virtual environment opens the door to promising new results on the geometrical foundations for potential "healing worlds" (Section 4.8).

The resulting synthesis of data from different disciplines presents a picture of the potential salutogenic (healing) benefits of virtual worlds, but only if they are designed to incorporate very specific geometrical and sensory characteristics. One possible surprise of our investigation is that simply porting architectural design from the physical to the virtual worlds does not necessarily accomplish the required goal, and we detail why this is so. It is mentioned here how tourist attractions—mainly of historical buildings and urban fabric—give clear hints on how to design healing environments (Section 4.9).

Section 4.10 suggests the need to apply design validation to the virtual environment. Surely, a world that people will spend a lot of time in needs to be assessed for its effects on human health. This methodology comes from the procedure for testing medical equipment and software, where a rigorous checklist is imposed by the U.S. Food and Drug Administration. Nothing like this exists in current architectural practice, while education does not prepare students to conduct any kind of scientific analysis.

Section 5 discusses the broad implications of beauty in virtual worlds, summarizing previous work on analyzing the design of the metaverse. We postulate the search for objective beauty in virtual worlds to be a key factor for users, especially as so many of them inhabit physical environments that lack the necessary emotional "background of visual beauty" almost entirely. Here we reference the philosophers Martin Heidegger and Gaston

Bachelard on the characteristics of what is perceived as "comfortable" settings, a space where humans can truly feel at home.

There is an enormous amount of recent research on what makes an environment comfortable to experience based on medical feedback and not aesthetics. Section 6 collects our ideas on which topics need to be pursued with research agendas. As the contribution comes from different effects, the effort needs to combine and coordinate multiple channels of investigation. Section 7 summarizes some psychological factors for appreciating beauty and applies the five-factor model to identify personality types that can profit from beauty in virtual worlds.

Artificial intelligence (AI) is evolving in parallel to virtual worlds, and some important findings on the appearance of objective beauty apply here (Section 8). We summarize recent applications of both verbal and visual AI programs to identify objective beauty, where the results point to what we already know from scientific analysis. Namely, that biologically based beauty is contained more in traditional design rather than industrial-modernist design. These results raise a warning flag to invite conventionally trained architects to design virtual worlds. At the same time, AI programs offer virtual testing tools for anticipating the attractiveness—hence the commercial success—of virtual worlds.

Section 9 covers the main points of biologically based beauty proposed as an essential background component of all healing environments. One major feature is the clear definition of the vertical axis needed to satisfy the body's gravitational reference. Even though verticality is often violated in the design of some buildings of the past few decades as a willful "transgressive" statement, doing so in virtual worlds is not recommended. Bringing in the concept of "cognitive capture" as one piece of an explanatory framework situates our suggestions in a broader mental model of how ideas are uncritically accepted by society.

Our conclusion, Section 10, summarizes ways in which designers of virtual worlds can determine whether they have a successful result in terms of what this article recommends.

3. Nature, Beauty, and Health

Extensive research shows that people find natural environments beautiful, and exposure to natural surroundings has been shown to exert a plethora of positive effects on mood [20], stress reduction [21], concentration and working memory [22–25], self-perceived health [26], self-esteem [27], recovery from surgery [6,28], as well as a reduction in criminal behavior [29,30]. Despite being an essential part of cultural folklore and inherited traditional wisdom, quantitative measurement and verification of the sensory effects of nature on health were only made possible by very recent technological developments. Neurological responses to environmental stimuli—in particular, visual ones—settle centuries-old questions of why natural settings are important for keeping the human body healthy.

These findings support the "biophilia hypothesis", which suggests that humans have an innate affinity for the natural environment and indicate that exposure to it can affect mental and physical health positively [28,31]. As humans evolved through being exposed to the geometry of nature, this mathematical/visual information field seems to have had a profound impact on the way that we connect to natural forms. "Structural" visual features of the environment can also be found in all pre-modern architecture, and those physical patterns are associated with the same positive health effects. An in-depth discussion about this "geometry of Nature" that permeates all natural and pre-modern artificial structures can be found in a number of publications and is beyond the scope of the present essay [32–36].

It is important for the present study that virtual environments mimicking natural ones reproduce the beneficial healing effects of physical environments [37–43]. Moreover, researchers are starting to use virtual environments to confirm and further investigate the impact that environmental geometry has on the users of architectural spaces [44–47]. These findings cannot be overemphasized, as they establish a link between healing and special informational properties—mainly visual—of the environment. That is, healing is triggered by special qualities that can be deliberately designed into both physical and virtual worlds. Design merges with medicine to inaugurate a vital new discipline.

4. Healing Beauty in Virtual Worlds

This section collects and lists distinct but related results from the gaming community. Virtual worlds have developed through an evolutionary process of trial-and-error, spurred by commercial considerations of market appeal. For the purposes of this paper, this discussion serves to underline the crucial separation of video games and the computer–user design interface (UX) from the standard design community coming from architecture and industrial products. Emotional/visceral appeal plays a crucial role in a successful process. This, in turn, is tied to objective beauty that arises from biological and mathematical criteria, not ephemeral fashion or intellectualization. While the data exist in the previous literature, our contribution is to synthesize them and draw strong conclusions from the synthesis.

4.1. Beauty in Video Game Worlds

In a study of what makes virtual worlds compelling and inviting to long-term play and interaction, Lee and Chen demonstrate that one key factor is aesthetic appraisal and that virtual worlds can be made more inviting by including beauty, variety, and breadth [48]. Importantly, visual beauty that features here corresponds to objective beauty that is related to a positive-valence visceral experience. This is entirely separate from intellectual appreciation or evaluation that could be totally subjective.

A good recent example is the video game *The Legend of Zelda: Tears of the Kingdom*, which was released in May 2023 to become the fastest-selling Nintendo game ever in the Americas [49] and see universal critical acclaim [50]. The game is noted for the immersion and beauty of the virtual world it portrays [51]. The Nintendo Switch platform shows the users how many hours were spent playing each game, and at this time of writing, fans are proudly sharing on social media how many hours they have spent playing *Zelda* [52].

4.2. Post-Avatar Depression Syndrome: Feeling down after Leaving the Theater

The *Avatar* film series directed by James Cameron is known for its stunning visual effects and, in particular, the lush, beautiful world that it portrays. When the first film was released in 2009, news outlets reported on "post-Avatar depression", or the extended feeling of gloom and dissatisfaction that persisted after they saw the film. The causes of this, in short, were the contrast between the pristine natural world depicted in the film in contrast with the soulless, developed physical world. Some viewers even reported suicidal ideation [53]. When the second film was released in 2022, such experiences surfaced again [54].

In analyzing what makes the world of Pandora (the habitable moon that is the setting for Avatar) so beautiful, commentators have pointed out that, to begin with, it depicts a natural rainforest environment replete with visual complexity and well-developed levels of scale (objects ranging from very tiny to mammoth), as well as ubiquitous curves and spiral forms. Beyond this, many of the plants are bioluminescent, creating multiple sources of soft light [55]. The fact that the plants do not just reflect light, as in a real-life scene, but actually project it makes them stand out even more and creates scenes with higher "doses" of natural geometry. This is a source of (*soft*) *fascination* which, together with *coherence*, provided by the organization described above, can make the scenes especially appealing [22].

4.3. The Metaphor of Architecture Applied to Digital Worlds

We *build* websites and software, not just make them. We say our websites and apps have software *architecture*, not structure in general. Such architectural terms are commonplace in software development and have been since the 1950s. As Halsted chronicles [56], such metaphors were initially adopted to emphasize a focus on the end-user of software; the needs and preferences of end-users were not much considered before then, but this metaphor eventually led to the field of human–computer interaction and the design paradigm of human-centered design. The metaphor of architecture also suggests a process of planning (that is, of design) in software development, occurring separately from and before the concrete work of building. Here is where the application of Alexander's design patterns creates a significant overlap between building architecture and software architecture [57], suggesting possibilities for future synergy. The point to note is that Alexander's patterns [58] generate adaptive environments whose principal feature is a high degree of "objective beauty", leading to psycho-physiological well-being [59]. The pattern language framework was ported into the computer world with great success in many distinct venues, not only software [60]. But we find it rather odd that the present-day effort to design virtual worlds so far neglects design patterns, despite their potential central role in shaping all types of environments. Design patterns tend to be used simply as a communicative tool in software engineering and interface design [61] rather than for their deeper purpose (in Alexander's estimation) of creating a living structure.

Kalay and Marx argued early on that it is necessary to design virtual worlds not as abstract spaces but as connected ensembles of places. The distinction is that *spaces* are impersonal and indifferent, whereas the geometry and informational content of *places* allow users to connect viscerally to them. (This also resonates with Heidegger's ideas about dwelling, discussed in Section 5 below.) People will seek a visual environment with which they can engage and experience from different viewpoints simultaneously, one that is adaptable, memorable, etc. [62]. The most successful environments, physical or virtual, have deep-seated attractive characteristics. Human neurological responses explain why the body unconsciously pushes a person to move in a particular direction, stimulated by the surrounding information field [63]. By contrast, an uninviting environment generates stress when one is forced to inhabit or move in it.

As previously mentioned, the appropriate design patterns channel a design in the direction of adaptivity to users. Connective design patterns for physical environments, for example, 2.4 *Biophilic Urbanism*, 6.1 *Place Network*, 12.1 *Handles*, 12.3 *Friendly Surfaces*, 15.2 *Human-Scale Detail*, 15.3 *Construction Ornament*, and 15.4 *Complex Materials* [59], also transfer to virtual worlds. The project of designing virtual worlds needs to figure out which tools from the physical world to emphasize and which to discard—virtual places are free of materiality and hence do not require tectonic elements ported from physical architecture [62]. Instead, the emphasis ought to shift almost entirely to connective placemaking techniques, now supported by neurophysiological data.

4.4. Flatness and Mindless Repetition, Threatening to Overtake the Digital World

A basic question for designers of virtual worlds turns out to be, how complex should it be? As will become clear from our discussion, objective beauty requires a certain threshold and organization of visual complexity; hence, it cannot arise in a monotonous environment.

To begin with a broader observation, notice how over the past decade, the term "content" has grown into an umbrella term that covers everything that can be found online, from academic research and newspaper articles to videos and artwork and even pornography [64]—as if all of these are the same kind of thing. At present, humans are just "content creators" [65]. One manifestation of such "flatness" is to impose a lack of variety, as if that somehow helps to achieve economy of scale. That notion is absurd, yet it persists in a visual typology of minimalism that eschews any variety. This, of course, creates an environment of sensory deprivation that is shunned by users, which is not exactly a useful prototype to apply to virtual worlds.

Previous work on monotonous repetition in visual environments is enough to condemn this design typology, even though it became an accepted standard during the 20th century. Physiological measurements reveal that monotonous repetition disturbs the human cognitive process and can lead to anxiety and headaches [66–69]. These results uncover a disturbing contradiction between the images of simplistic modularity linked to crude industrial production and actual human reactions to built structures that embody those images. Despite the documented evidence, it has proved impossible to convince the architectural profession to drop this noxious building typology from practice. The early 20th-century myth linking the *appearance* of efficiency (unrelated to the real efficiency of any process) by means of monotonous repetition is stuck in the collective consciousness [70].

This historical background explains why the look of most buildings in virtual worlds follows these typologies. They have become the default building designs, and VR creators transfer them into virtual worlds without considering what purpose those typologies would serve there. The deliberate construction of unfriendly environments is a special case, of course, but what is remarkable is that VR designers often create such environments unwittingly by following the current norm in the physical world.

4.5. On Beauty Being Absent from Considerations of Digital Design

An everyday understanding of "good design" (mostly among non-architects and designers) almost certainly includes some reference to beauty. In the technology field, however, the common understanding of "good design" has long been completely separate from beauty. For example, in Gould and Lewis' seminal article defining the tenets of user-centered design, no mention is made of beauty [71]. For them, rather, good design is about a focus on users, empirical measurement, and iterative improvement. Noted design theorist Donald Norman (who coined the term "user experience" in 1993) even suggested in his classic *The Design of Everyday Things* that "beauty" is used to cover up design flaws related to usability [72].

To be sure, other digital designers have examined the role of beauty in digital design, discovering links between beauty and usability. The academic results have been nuanced and mixed in certain respects, yet there has been a discernible "aesthetic–usability effect" which states that users perceive aesthetically pleasing designs as more usable [73]. Alexander argues that, in good design, beauty and functionality both spring from the same source. It is not a question of either/or. In the best designs, the different elements help and strengthen each other, and in so doing, the whole also becomes more beautiful.

Why are digital designers not anchored on and driven by beauty? Perhaps because our schools do not teach how to create beauty anymore [74,75]. As the dominant culture does not produce environments with visceral beauty that trigger a nourishing physiological state, the methods for producing it remain a mystery to most people. Where is a young designer who has a gut feeling for beauty going to learn its basic principles to apply to their work? (So far, Alexander's books are not standard texts in Art and Architecture courses.) But the problem goes far deeper: the collective narrative falsely proclaims that a logical basis for beauty does not exist; that the notion is totally subjective. Defining objective beauty becomes a terrifying prospect for certain people.

Another reason for this neglect may be that digital design has its roots predominantly in computer science, and while this field may lend itself to beauty in the sense of elegant mathematical solutions and beautiful programming languages, it has not historically explored beauty in terms of the experiences end-users have with the software. As the field of user experience has grown over the past two decades and taken hold as a profession unto itself, education and research on digital design are beginning to acknowledge beauty, but this is still nascent. For example, the major textbook in the field, *Interaction Design*, by Sharp, Rogers, and Preece [76], whose sixth edition has just been published at the time of this writing, makes virtually no mention of beauty or aesthetics. The exception is a single paragraph (p. 182) mentioning the aesthetic–usability effect discussed above.

A deeper reason may lie in the historical, philosophical trend toward "objective" knowledge, which at some point began to exclude experiences like beauty as "subjective". This trend has its roots in 17th-century philosophers like René Descartes, for whom subjective knowledge was seen as "unscientific" and, therefore, to be ignored. Indeed, around the time of Descartes, Sir Henry Wotton famously translated the "Vitruvian triad" of firmitas, utilitas, venustas, as "firmness, commodity, delight"—even though the literal meaning of the Latin word "venustas" is indeed "beauty". This was a revealing edit.

More recently, subjective experiences have famously re-entered the scientific purview through fields like neuroscience, psychology, and evolutionary biology. Even mathematicians have been able to describe the various properties that are consistently perceived as beautiful, including

specific classes of symmetric and fractal structures [33–36,77]. Alexander reveals that objective beauty is anchored on scientific truth, a conclusion reinforced by the experiments referenced here [18,35]. To help ensure that a virtual world is restorative, designers are recommended to search for the inherent/latent/unrealized beauty unsullied by the pursuit of fashion.

4.6. Beauty Defined for Physical Architecture, Then Ported into Virtual Architecture

A unifying theme of this essay is transferring visual environments from physical to virtual reality. Inexplicably, this has not generally been approached as a classic problem in porting a user interface across platforms (for which techniques are known). Despite the commercial importance of doing it right, developers naïvely assume that dominant architectural styles are fit for virtual worlds without extensive modifications. The process is compromised, however, by (1) porting tectonic elements from physical architecture that are irrelevant and unnecessary in a virtual context and (2) porting design elements that discourage users in either platform.

We previously undertook a research project to document the essential geometrical ideas underpinning objective beauty [34,66,70,77–80]. This project follows the notion of beauty as a necessary organic constituent of the human environment, introduced by Christopher Alexander [18,35,81–83] and, in parallel, by the philosopher Sir Roger Scruton and others [84]. The current traditional architects always focus on objective beauty [85,86]. We argue that the widely perceived failure of objective beauty in virtual worlds was due to ignorance of the methods for producing it and propose developing a pattern language for designing the metaverse [87]. Professionals working to create an attractive virtual world simply copied the dominant aesthetic from physical buildings without realizing that it does not lead to visceral beauty but often its opposite [88]. As further support for this conclusion, consider how the physical architecture of the clustered tech company headquarters that are involved in virtual world innovation in the US is surprisingly unattractive [89]. Even though design rules for producing a wonderfully health-enhancing environment are known, they were ignored in favor of banal modernist or 'fashionable' building designs.

We have also previously discussed the success of the Roku City screensaver [87]. The presence of rows of coherent and detailed human-scale façades in this environment, ranging from Art-Deco to Neoclassical, represents an area that users want to inhabit, despite the presence of some unnatural features, like the purple-blue light that bathes the scene. This demonstrates that it is not the exact replication of an existing, real-world scene but the incorporation of specific types of geometrical organization that makes this VR environment attractive. In addition, the extensive natural park and open water that separates this area from a dense built-up city seen in the far distance, where monsters, pirates, and even an erupting volcano can be found, create an archetypal prospect-and-refuge scenario, where the coherent environment represents a refuge, from where potential threats, situated in an industrial-modernist environment, can be monitored from a safe distance [90,91].

Here, we see that multiple levels of perception and reaction patterns, deeply rooted in our very distant past, are played out in a digital environment not explicitly created to utilize them. Instead, this resonance is brought about by a visual vocabulary shared viscerally by the designer and the users.

4.7. Apple Vision Pro Design Guidelines

In June 2023, Apple announced its upcoming line of mixed-reality headsets, the first product of which, the Vision Pro, will be released sometime in 2024. As a mixed-reality product, the device allows users to undergo fully immersive experiences in virtual worlds (virtual reality) as well as experiences in which digital elements are overlaid on the user's real physical environment (augmented reality).

As with all their platforms, Apple has released a set of guidelines for designing its Vision products [92]. Throughout these guidelines, an emphasis is placed on creating a beautiful and comfortable experience for users. Some of the specific guidelines include leveraging space and naturalistic gestures so that users "feel at home on the device", using a

"glass" background whose translucent appearance adapts to its physical surroundings [93], and designing objects (including icons) with multiple layers to include complexity, shading, and depth, like objects in the real world [94].

4.8. Virtual Therapeutic Environments

Virtual worlds can be designed to offer a therapeutic environment by applying the appropriate qualities [95,96]. Changing the actual physical environment to enhance informational health benefits is strictly limited, with practical options being restricted to altering the color illumination or the piped music. It is not feasible to change room dimensions, physical spaces, surfaces, the placement of doors and windows, etc. A virtual world, however, has unlimited freedom to introduce changes in real time and to assess their effects on users through sensors. State-of-the-art wearable sensors recognize physiological changes due to positive emotions, thus documenting the user's state of well-being [97,98]. This possibility of easy experiments performed by adjusting a virtual environment to regulate users' emotions towards a positive mood offers enormous opportunities for improving human health.

Creating adaptive virtual environments that respond dynamically to the users' perceptions in real time opens the door to a new era of research [99]. A mechanism based on neuroscience data can exploit such a feedback loop for investigating the geometry of healing environments. A physical and virtual design can use design characteristics associated with a positive state while new results continue to augment the applied knowledge base. This research will add to and complement documented design patterns that are discovered by analytical observation, and new results could be formulated in terms of new patterns.

A study that manipulated architectural variables in virtual reality shows how this could be performed. The shape, size, color, and curved walls of a virtual space were changed while observers recorded their responses to the changes [100]. The authors of that study concluded that human aesthetic experience could be reduced to the three factors of familiarity, excitement, and fascination, partially overlapping with findings using images of interiors where fascination, coherence, and hominess were identified [101]. A different study varied the characteristics of indoor virtual environments (e.g., ceiling height, wall color, floor material) and recorded the users' responses [102].

The COVID-19 pandemic gave an incentive for research into the healing effects of visual input. Several independent investigators reached the same conclusions on the characteristics of a virtual healing environment—fractal shapes, color, plants, natural scenery, etc. [103]. The design of simulated places is based on existing computer games that utilize these factors to create attractive settings. Yet another application of VR simulations reduces chronic pain in a broad spectrum of cases [104]. Again, this simulation is based on creating an engaging, pleasant, and rich visual experience. A very different VR application employs realistic scenery to help combat veterans suffering from PTSD (post-traumatic stress disorder) go through healing exercises [105]. Therefore, diversified VR healing applications can all be attributed to the specific geometry of "living structure".

4.9. Virtual Worlds Competing with Tourist Attractions

International tourism provides a time-tested precedent for environments that attract users. People happily spend money to travel to a location, then spend more money on activities there if that spot is viscerally attractive. Tourist destinations include locations of natural beauty and built beauty embedded in traditional architecture and urbanism. Those informational environments lure people out of their own everyday settings with the promise of experiencing vastly augmented beauty. The aesthetic aspect, rather than the historical context that is always touted, is what attracts users [106,107]. Recent studies rank the restorative sensory effect from historical architecture to be comparable to that obtained from natural scenery [108].

Creators of virtual worlds can learn valuable lessons from the tourism industry. Sensorial qualities common to natural and traditional architectural environments are the opposite of industrial-modernist, minimalist, or Deconstructivist structures [18,35,36,63,70,77–83,86]. Without wandering off-topic, we mention that Disney World created a highly successful model of a new simulacrum or emulation—in a sense, a good example of a "physical virtual" world [109]. A lot has been written about Disney undertaking original research on the elements contributing to viscerally attractive environments while ignoring the advice of mainstream architects [110].

4.10. Contradictions with Design Education

Starting from the Bauhaus's central ideological aim, architectural education erased historical building precedents in a radical wave of innovation. Identifiable historical content in new buildings was explicitly condemned starting from the beginning of the 20th century up until post-modernism, after which historical references were allowed only as sarcastic jokes, not as integral components—with the exception perhaps of British postmodernism [111]. Dominant architectural culture ignores the few architects practicing a traditional idiom as not being part of the mainstream [86,112–116]. Since mainstream opinion forbids using traditional techniques to build at present, students are not taught how to create places with the same visceral attraction as historical ones and are, moreover, morally forbidden to do so. This rigid restriction leaves little room for the injection of objective beauty into virtual worlds, which can consequently be achieved only by sidestepping dominant architectural culture.

Readers of this essay, such as developers and users of gaming and virtual worlds, might imagine that our warning against engaging architects is exaggerated. Nevertheless, the disciplines split because of inherently distinct methods of design validation. An essential prerequisite for adaptive design is to collect and synthesize factors that affect users' unconscious behavior in real and virtual environments, which is what is attempted here. But note that our references come from the scientific literature—all of them outside architectural academia. The documented research contrasts starkly with current architectural thinking [87]. This striking disparity underlines the little-known fact that architecture students are not trained to do research, do not know the fundamentals of critical investigation and scientific reasoning, and misuse the term "research" to describe the results of artistic variation [36,70,117]. Despite lacking the indispensable skillsets of adaptive design validation [118], architects present themselves as being uniquely qualified to design virtual worlds [119].

VR developers should watch out for architects' misuse of scientific words to sell their designs [120–122]. Architectural texts can demonstrate great facility in appropriating and manipulating terminology from neuroscience, but ultimately their purpose is to promote either their author or some contemporary practitioner [123]. This is neither science, as it is not based on actual data, nor academic scholarship, but calculated marketing. Such discourse is confusing because it is dense with jargon—it almost makes sense but ultimately lacks logical consistency, like a bad AI program that produces plausible but nonsensical text [124]. The product (a photo of a building or rendering of a proposal) gives the game away by contradicting neuro-design principles, despite claiming the opposite.

5. Beauty in the Metaverse

An attractive virtual world has to provide an escape from the physical one. If not, then why enter it? Certainly, the activities offered to the user in the virtual environment are a major—perhaps the major—motivating factor, but the attractiveness of the virtual environment must also come into play. Given the deteriorating aesthetics on the ground in many parts of the globe, a virtual world will likely appeal to individuals whose physical environment looks severely industrial/modernist (this conjecture poses an interesting experiment that can be performed with existing online tools). Life in such surroundings is not very appealing, as perceived unconsciously by the human body, regardless of whatever aesthetics happen to be fashionable (and are imposed top-down by a heavy-handed construction industry).

In East Asia, millions of users of virtual worlds are drawn from among a population residing and working in dystopian, inhumane mass housing [125]. Indeed, the popularity of video games prompted the Chinese government to implement limits on kids' game time [126]. Dated "futuristic" architecture (from the 1920s, reinforced by Disney's 1955

Tomorrowland) is precisely the sort of unresponsive environment users are trying to sneak away from. Implementing "virtual reality greenspaces" provides a valuable measure to help people cope with stress due to chronic mental fatigue experienced in over-urbanized cities. By promoting attention restoration, VR exposure as a surrogate to nature may turn into a beneficial tool for improving health [127].

At the same time, the prospect of escaping physical reality—along with the responsibilities for negotiating the real world—for a virtual alternative can grow into a serious problem. We accuse architects and planners of having made the physical built environment so dystopic and flawed that being able to escape from it seems like an attractive idea [87]. Our long-term hope, however, is for this escape to trigger a massive reconsideration of how to create an objectively beautiful physical environment. The impetus will hopefully come from profit-oriented forces trying to draw virtual users back into reality, which will then help the world to implement objective beauty.

Many of our proposals resonate with VR developers who have heuristically discovered that users are attracted to virtual worlds. Chloe Sun lists a variety of generative art software that can be combined to create a more "living" virtual environment inspired by Alexander's foundational work [128,129]. Outside the insular world of architecture and art establishments, game designers are free to investigate evidence-based and intuition-driven design solutions [130,131]. Bringing these ideas together into an epistemological framework for VR helps define a design paradigm that is consistent with human neurological response.

The important point is that design patterns (both Alexandrian and software) did not originally focus on aesthetics, beauty, geometry, etc. but focused instead on "function" (i.e., forces, processes, etc.). The deeper sense of beauty was the topic of the more philosophical book The Timeless Way of Building, in which Alexander introduces the QWAN—the "quality without a name" [132–134]. To this day, Alexander's A Pattern Language is well-known and cited among computer scientists (though usually only in passing, referring to the "function" aspect of design patterns), while The Timeless Way of Building is much less cited. Still, computer scientists who recognize and value elegant software—beautiful programs—do reference the QWAN [135]. It is worth noting that Alexander's own understanding of the patterns greatly expanded later, during his writing of The Nature of Order, to embrace geometrical transformations and the emergence of beauty through iteration and growth, what he called "unfolding" [18,35]. This suggests certain resonances with software development, which is made iteratively and increasingly through the processes of human-centered design and with attention to broader aspects of human experience (dubbed user experience or UX) and not just narrow usability. It seems that the software people have been playing catch-up, but the ones programming the metaverse seem to have been completely left behind!

Is it possible for the metaverse to be a healing environment? Even before getting into Christopher Alexander's work, we can refer to Martin Heidegger's ideas of dwelling in space [136]. As Heidegger argued, "dwelling" is about being at home in one's totality—physically, psychologically, physiologically, and spiritually. At his time of writing in the mid-1940s, the dominant architectural understanding of "dwelling" was simply having a place to store one's physical body. But with his more holistic understanding of dwelling, Heidegger argued that for humans to feel truly at home in a place, it must have certain characteristics: a supporting ground, which he called Earth; a sense of something beyond (Sky); reminders for us of our own mortality, which helps us make the most of life (Mortals); and meaningful objects and experiences that we can undergo (Divinities). Environments, whether physical or virtual, that do not integrate these four elements, Heidegger suggests, will not inspire users to care for the environment or strive for meaningful goals, let alone feel comfortable or satisfied [137].

On the topic of how philosophers view habitable space, Gaston Bachelard described a desirable state of psychological well-being as follows: "I should say: the house shelters daydreaming, the house protects the dreamer, the house allows one to dream in peace." [138]. We interpret this condition as a state of the body free from anxiety induced by environmental stress and especially lacking cognitive stress due to hostile elements of the geometry. This is the

type of calming and effortless architectural experience that is conducive to human health and which invites us to explore a virtual world as if in a dream (but not a nightmare!).

Philosophical investigations on the sense of effortless belonging to places contextualize the question excellently and set a strong foundation for future health-centric empirical assessments of design features.

6. Opportunities for Further Research

While the metaverse took off with little understanding of underlying factors, that hype has subsided. Now that Apple has announced its upcoming headset for 2024, there is a renewed interest in designing for virtual worlds. It is fascinating to read their design guidelines (see Section 4.7 above). Following our analysis of the situation, it seems to the present authors that Apple is at the forefront of communicating what it could mean to design for a human being in VR.

The existing literature covers known impacts of online experience on health and wellbeing, which is mostly focused on social relationships. To complement this, we wish to investigate the primary effects of the environment itself on human well-being due to its colors, geometry, shapes, spaces, surfaces, etc. Even if we focus strictly on the presence and different types of complexity that incorporates symmetry, there are key results that affect health. Our work on symmetry helps to document its relationship to salutogenic structures and their opposites [77,139–141].

We understand the extensive challenges involved in conducting rigorous research in this area. Yet we urgently need to see a method of analysis developed further that resolves existing conundrums. One of these is a contradiction in terms where contemporary architects love their designs and characterize them as "beautiful", whereas they are not attractive to others. Future research should help to distinguish idiosyncratic from objective perceptions of beauty.

A fascinating perspective has been added by the finding, with functional MRI, that the processing of recursive (fractal) forms recruits different resources from the processing of nonrecursive forms. Remarkably, this process recruits the Default Mode Network, a functional brain network known to be involved in the processing of internal information across multiple forms of complex cognition, many of which are linked to memory or abstract thought [142,143]. This implies that such forms (clearly showing multiple nested symmetries interlinked by scaling symmetries) are higher in perceptual fluency, and do not require specific task-oriented networks for their processing, something that fits well with the concept of "soft fascination", i.e., the involuntary effortless attention induced by natural stimuli, as opposed to "directed attention", i.e., the voluntary use of attention for task execution.

Some interesting initial papers relate to the physiological effects of visual stimuli in non-virtual worlds [144,145]. The question is: how to use those results obtained from the physical world to better design virtual worlds? But then this definition raises more questions than it answers. However, they are the RIGHT questions by asking WHO is doing the designing, in what context, and HOW do we know what the right (effective) courses of action are? How do we really know what th

e existing conditions are (since these are complex and all knowledge of them is incomplete)? And especially, how do we know what conditions users really prefer? What happens when those change?

Many young people (and increasingly others) spend countless hours in virtual game environments already. A recent survey showed that teens spend an average of two hours a day playing video games! So, one need not buy all the boosterish speculation about the virtual future to have reason to be very concerned about the existing (and growing) impact on mental and physical health.

This paper shifts the emphasis a little bit. The idea is that there is an unmet professional responsibility to make actual healing environments, whether they be physical or virtual. So far, the architect's innovation has been given unchecked free rein, sometimes at the expense of human health. We are not giving the design professionals "a pass" so to speak. Instead, we focus on the fact that so many billions of people spend increasing amounts of time

online and that in itself poses a grave health threat [146]. Therefore, there is a professional responsibility to make virtual worlds into healing environments, to the degree that this is feasible, by employing the best of our relevant scientific knowledge and skills.

The next step is the more important one: we can apply those lessons learned from virtual worlds to shape the physical world and upgrade its ugly spots. This is an example of "connecting the dots" to existing research. We must deal with the impacts of aesthetics on health and well-being, with a focus on brain function (e.g., the important work of Semir Zeki and closely related research) [147–149]. Cleo Valentine's work on neuroinflammation defines new and important directions in this investigation [150].

7. Neurological Appreciation of Beauty: Insights from the Five-Factor Model in Psychology

A deeper understanding of who uses virtual worlds and why relies on discovering what motivates the users. Clearly, a need to search for emotional experiences not provided by the physical world combines with the spirit of exploration to drive users. This drive is contingent upon particular character traits that vary among individuals.

To identify the personality profile that is associated more closely with this drive, we must look at personality traits. The most influential and widely used personality model in Psychology is the five-factor model (FFM). The FFM creates a personality profile based on the individual's position along five basic dimensions or traits, which have been identified by factor analysis: (a) Extraversion, (b) Agreeableness, (c) Conscientiousness, (d) Neuroticism, and (e) Openness to Experience [151]. Each of these major traits can be further analyzed into facets to give a higher-resolution analysis of someone's personality when that is required.

Costa and McCrae identified a more specific quality, *Openness to aesthetics*, or the trait *Appreciation of Beauty (tAoB)*, as a specific subtrait within the broader trait of Openness to Experience and described it as "a deep appreciation for art and beauty" [152,153]. *TAoB* was found to have the highest correlation to the trait of Openness than any other subtrait, indicating its significance and, potentially, its prototypical role in the overall trait of Openness.

Appreciation of beauty has also been positively correlated with traits (b) Agreeableness and (a) Extraversion [154]. By using factor analysis, Diessner et al. determined that there are, in turn, at least three sub-traits of tAoB (called Engagement with Beauty by these authors): (1) the trait of appreciating natural beauty, (2) the trait of appreciating artistic beauty (including Architecture), and (3) the trait of appreciating moral beauty [155].

A new study about users of the ZEPETO South Korean mobile chat app, where users create and interact as 3D avatars in different worlds, has revealed some interesting results about their psychological profile [156]. Zepeto users are higher on Extraversion and Openness than non-users; overall, they are driven by the desire to explore the virtual world and enjoy unique experiences, but such a tendency was stronger among those higher in the traits of Openness and Agreeableness, while those high in the trait Neuroticism were seeking more of an escape from reality.

In another recent study, a meta-analysis examining general social media use, Extroversion and Openness were significantly and consistently related to the use of such platforms [157]. Internet addiction, on the other hand, correlates with different personality profiles; Neuroticism has been shown to be positively correlated with internet addiction, whereas Openness, Agreeableness, and Conscientiousness are negatively correlated [158].

When taken together, these results illustrate an interesting finding: (non-addicted) users of the metaverse, and social media in general, seem to score higher in exactly those traits that make them more likely to appreciate/engage with beauty. This correlation points us in the direction of a potential opportunity: these users, when exposed to beauty in a virtual world, are more likely to benefit from the aesthetically rewarding stimulus and, presumably, will seek to translate this benefit into the real world.

8. AI Can Discover Important New Results in Adaptive Design

Artificial intelligence (AI) is often discussed in the same context as virtual worlds. The development of AI opens exciting possibilities by providing tools with which to explore adaptive (which encompasses human-centered) design [159]. Multiple factors defined by neurobiology and physiology determine whether an environment can be used comfortably—or be avoided. The same human-centered considerations apply to virtual worlds, albeit in a somewhat restricted spectrum. The ensemble of human senses that play such a determining role in perceiving physical environments is reduced to the visual (and auditory) realms in virtual worlds. Because of the overwhelming preponderance of visual information in sensory stimulation that adapts humans to their environment, this restriction is not a major issue. We are not directly measuring what is known as "hedonic quality", referring to a positive subjective experience. Rather, we are referring to a state of positive health and well-being, which is an *objective* physio-psychological state that correlates with comfort and reduced stress.

So far, applications of AI to building design focus on generating abstract images unrelated to human adaptation; that is, creating a design that is visually attractive as a distant image, model, or object but whose suitability for human habitation and use is left unexplored [160]. We would like to draw attention to the totally distinct approaches to design practiced by adaptive versus conventional architects. Designing according to human sensibilities grounded upon biomedical data begins with the body's responses to colors, details, shapes, surfaces, symmetries, and volumes. The forms of environmental structures, whether representing a physical or virtual environment, arise from those adaptations. This process implies that nothing is arbitrarily imposed top-down. By contrast, the obvious human-centered considerations play little or no role in contemporary design as practiced by the dominant architectural culture. What we see instead is AI being used to generate random forms that look "innovative" and are destined to be built with standard global industrial-modernist materials. The design of virtual worlds follows this model whenever architects get involved.

However, any approach that tends to focus excessively on innovation for the sake of innovation, overlooking deep/systemic issues, etc., should be avoided. Lately, people have been using the alternative term "humanity-centered design" to try to capture more of an Alexandrian approach to this topic. The established profession is not interested in applying AI to explore design adaptivity to users. The 20th-century dominant architectural culture was never concerned with sensory feedback. Starting with early modernism, architectural culture has adhered to a top-down model that creates an image of a building, then introduces it into the world as a built "object". The decision-making process of generating physical forms in the environment as untested abstractions has been hugely successful for businesses and the building industry. Just like in the physical world, the same result in virtual worlds is insensitive to the physio-psychological factors governing human use, without, in this case, having the financial advantage that it has in the physical world.

AI tools make it possible to investigate the precise qualities that define an adaptive environment that enhances human health. Both verbal (ChatGPT) [161,162] and visual (Craiyon, DALL·E 2, DreamStudio, Midjourney, Stable Diffusion) [163] applications of AI can, and should, be used to explore what factors make an environment attractive and comfortable to experience. This program towards validation goes beyond any flashy attraction by privileging medically validated positive factors for well-being. The massive data bank that AI draws upon to either create imaginary images or to compose a verbal description includes billions of pieces of human data. AI applied to conduct research is akin to an enormously scaled-up survey of people's unconscious preferences.

The numbers win out—and we have already discussed how such applications always produce "traditional"-looking buildings (in the historical timeline up to and including Art Deco) when prompted to create a beautiful building image; versus industrial-modernist building images when prompted for ugliness. These results are consistent among themselves, as they also conform with biophilia. Visual AI generators identify as ugly those environments that dominant architectural culture privileges as paradigmatic and which it impulsively transfers into virtual worlds. A disturbing question posed in this essay's Introduction—whether AI would generate dystopian virtual worlds—depends upon whether developers choose to enlist AI to design and validate healing environments (which look surprisingly different from the images that dominate architectural training). Visual AI is a powerful and useful tool that can generate either healing or uncomfortable designs. Society cannot relinquish this determining choice to those with a stylistic agenda. Moreover, the possible future development of AI suggests possibilities for creating nourishing virtual worlds beyond the human imagination; rather than appearing to us as "naturalistic", such virtual worlds might strike us as "otherworldly" but nonetheless comforting and homelike.

Adaptive architecture focuses on human needs from the immediate point of view of the user. It confronts and must solve a myriad of technical problems such as the optimal use of local materials; energy-savings through climate adaptation; integration of the building with nature; how to fit a building into the existing urban fabric; adjustment of spaces for human movement and psychological well-being; positioning of entrances and windows; designing unconsciously welcoming surfaces, etc. Many of these design constraints—particularly those that affect human senses—are shared with virtual worlds. This approach makes it easy to anticipate the valence (negative or positive) of the resulting user interaction.

Applications of both verbal and visual AI overwhelmingly verify what we already know from biophilia, design coherence, emotional affordances, fractals, symmetries, visual attention scans, etc. [79–83,87] The software, in its various applications, simply validates the informational content that is embedded in traditional and vernacular architecture. Beauty in nature arises when form merges with function, which transfers into the exceptional qualities of traditional architecture to generate well-being. The human-centered design avoids the established "canonical" building typologies that dominate current official construction since those represent a rather narrow establishment opinion: that validates designs exclusively by how faithfully they conform to a peculiar aesthetic [36,70,120,164].

9. Discussion: Escaping Cognitive Capture

This study argued for the importance of aesthetic experience in designing virtual worlds. It was important to define a particular type of aesthetic—biologically based objective beauty—to which the human body responds positively, though unconsciously. Bringing together previous results by many researchers, including the present authors, creates a synthesis of ideas about what people respond to viscerally. The design profession finds itself at a turning point when knowledge from mathematics, medicine, neuroscience, and psychology is rich enough to have predictive value. Yet this approach to design is still non-standard, being largely unknown to mainstream architects of either physical or virtual worlds.

The reasons for this state of affairs are beyond the scope of this paper, yet we can observe several important factors. What is known as "cognitive capture" [165], in which a population begins to take on the modes of thought of an industry or profession, results in a diminished capacity for independent or critical thought. Such a capacity is diminished because the population is distracted, or "blinded", by attention-absorbing features or practices, and their normal processes of discernment and judgment are allowed to atrophy. A related phenomenon is known as "architectural myopia" [166], wherein architects have been observed to make aesthetic judgments that depart from the judgments of most non-architects, in part because they are trained to favor aesthetics that conform to simplistic industrial geometries (a habit of thought described by the present authors as "geometrical fundamentalism") [19,70,87].

A similar problem can be seen arising in the computer science industry. The more primitive geometries of earlier graphics systems have come to dominate later graphics and even become popular—another instance of "cognitive capture", this time of the users. The problem is compounded by the atrophy of users' capacities for fine aesthetic judgments as they become

passive consumers of ever more random, titillating, but ultimately non-nourishing aesthetic experiences. For children, this represents a profound threat not only to their health but to their capacity to seek salutogenic qualities later in life and to participate in a democracy (and, in some cases, as professionals) in building more salutogenic environments.

There is, however, an alternative to this slow erosion of aesthetic quality and user capacities. As we have outlined here, verified adaptive design rules can now be applied with confidence to create virtual worlds that are not only appealing in their sensory effect but are healing in the long term and also afford the ability to develop capacities for richer aesthetic discernments and explorations [87]. We, therefore, urge developers to shape virtual worlds in the same way as giving free rein for children to color and ornament their immediate environment (before their inborn creativity is quashed by a dreary forced aesthetic). Having to inhabit the result puts a brake on dangerous fancies. Ordinary users will naturally be attracted preferentially to those mostly visual characteristics without "knowing" why (because the attraction is unconscious). No matter. A truly revolutionary development occurs if we can improve the health of users through the design of the virtual worlds in which they are immersed.

The issue of the incomplete nature of the virtual experience can be seen as a potential obstacle and has already been discussed [4,150]. A serious limitation of VR is the lack of gravitational sensation (and generally accelerational sensation, except in specialized installations) and, hence, the lack of multisensory integration. There is no doubt that these differentiate the experience of VR from the real thing, and this is why it can be argued that, in studies where simple reactions to images are investigated, there are some cases where it is better to show photos on a computer screen than to use headsets; because with the screen, we see what we expect, for example, a 2D image on a surface, without the perceptual dissonance of the VR headset.

Note that an unambiguous definition of the gravitational axis in the design of a building (both its interiors and outside appearance) is one of the factors of the "Biophilic Healing Index" [10]. Lacking an explicit vertical reference, or a virtual vertical axis defined through bilateral symmetry, such as an arch, creates problems in balance. We are referring here to a static effect due to the perception of environmental geometry and not to motion-induced nausea. Without getting into a discussion of this effect in physical settings, the analogous lack of a clear vertical reference is one factor that results in cybersickness [167,168].

Because of these problems, it has been argued that VR studies may be insufficient to fully capture the consequences of extended exposure on human health. But this does not bear on our discussion here. New studies are needed to investigate the effects of changing just one parameter, i.e., the aspects of the geometries presented, one at a time. All other parameters—exposure time, properties of the headsets and the operating systems, etc.—should be identical. From an experimental point of view, this would be the way forward. From a conceptual point of view, none of these concerns goes against our argument about the positive effects of incorporating truly biophilic design in the metaverse, all things being equal, regardless of the potential negative effects of spending time in the metaverse.

10. Conclusions

With the ever-increasing use of VR platforms, we now have at our disposal a whole new realm—whole new worlds, we might say—in which principles of human-centered, health-promoting design can be applied. At the same time, we are confronted with growing evidence of the health impacts of our virtual-world designs, both positive and negative—carrying normative ethical implications, as we have explored.

Design in these worlds is not governed by financial constraints, unlike the physical world of buildings and landscapes, and there is no reason, other than neglect, habit, reckless profit-seeking, or ideologically driven imposition, that these environments should not be built in the best human-centered way possible. Indeed, if designers see themselves as true professionals, then they surely have an ethical duty of care to the well-being of those for whom they design. In this paper, we have proposed a normative standard for designers to provide a universal visual experience arising from a persistent background of biologically based objective beauty and affording the capacity of exploration, restorative play, and development of life skills in aesthetic discernment.

As we have argued in previous publications, in contrast to almost all of human history, the most recent century of design can best be characterized as a project to apply distracting fig leaves to a crude and unsustainable phase of industrialization: doubling down on "geometrical fundamentalism", industrial cognitive capture, and financially manipulative "art stunts". At a low level, the resulting aesthetic experiences are harmless carnival fun. However, evidence is mounting that, when consumed as a steady diet—as they are increasingly by people of all ages, and especially the young—such experiences are having a profound pathogenic effect, with long-term negative implications for a free and democratic society.

To be sure, virtual worlds can be expected to include ugliness, surprise, stress, and other painful aspects of life. These experiences can also be therapeutic when they are presented in transitory modes or from a sufficient perspective. The question for designers is whether the background environments are barren, ugly, overly mechanical, or otherwise degrading of aesthetic experiences—or conversely, whether they are restorative, naturalistic, richly complex—in a word, beautiful. A parallel question is whether users are passive consumers to be manipulated, titillated, and exploited or whether they are actively engaged in exploring (and creating) aesthetic richness.

The core argument is that designers of virtual worlds should focus on creating worlds that are, for the most part, restorative, naturalistic, engaging, and richly complex. Our suggestions to help designers do this boil down to two complementary approaches. One of them involves the technical application of design techniques abstracted from biophilia, fractals, nested symmetries, Alexander's 15 geometrical properties, etc., which are discussed in depth in our other work, as well as by other authors. Even though these methods do not yet constitute a standard part of mainstream art and design, their details are already spelled out in the literature.

The second approach is to trust one's intuitive and visceral feeling of objective beauty and not be led astray by existing agendas coming from the design profession. This method is far more direct but paradoxically more difficult to apply because of unconscious social pressure and the atrophy of healthy aesthetic discernment skills as a result of cognitive capture. Yet as Christopher Alexander and others have argued, it is possible to re-train one's body to liberate the child-like positive reaction to objective beauty that is everyone's evolutionary heritage. This could be facilitated in practice by employing wearable sensors that measure bodily responses (such as activation level, mood, and stress level), which resolve cognitive dissonance triggered by prevalent media opinions. Our discussion gave some practical guidelines for sharpening the appreciation of beautiful environments by paying attention to neurophysiology.

The way in which designers know they have achieved a successful result in terms of user emotions is precisely by what the users "feel". Over the long term, it draws users back for reasons they cannot clearly explain—it offers a comfortable, easy, emotionally intuitive fit. This unconscious reaction (related to Alexander's QWAN—quality without a name, see Section 5 above) might not be easy to describe, and for that reason, researchers and designers may be hesitant to attempt to operationalize it. We can ensure that a virtual world is restorative to a significant extent by how far it satisfies the geometrical criteria that link to neurophysiology and which come from human evolution. Of course, there will always be additional factors that are beyond our control. A virtual world need not be naturalistic—yet its geometry must satisfy the requisite mathematical properties. There is a place for beautiful, otherworldly places that succeed because they satisfy those properties, as we find in successful gaming environments. A scene "feels" real but does not necessarily duplicate nature. Note also that an emotionally successful setting needs to exceed a complexity threshold; otherwise, the requisite mathematical properties cannot be satisfied. If one looks carefully at a beautiful plain artifact or setting, it may indeed look very sparse

at first glance; however, one will invariably discover unnoticed layers of complexity that support its overall coherence.

We are optimistic about the opportunities, including further research progress. As one example, the recent advances of AI mean that the collection and integration of data from user experience in the virtual worlds (both positive and negative, as much of the virtual environment can be deliberately unfriendly in games) can be used to further promote human-centered design in the physical world. We are now standing at a new crossroads in the history of environmental design, and this is an opportunity—and perhaps a duty—that we should not pass by.

Author Contributions: Conceptualization, T.G., N.A.S., A.A.L. and M.W.M.; methodology, T.G., N.A.S., A.A.L. and M.W.M.; writing—original draft preparation, N.A.S.; writing—review and editing, T.G., N.A.S., A.A.L. and M.W.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Vivolo-Kantor, A.M.; Martell, B.N.; Holland, K.M.; Westby, R. A systematic review and content analysis of bullying and cyber-bullying measurement strategies. *Aggress. Violent Behav.* **2014**, *19*, 423–434. [CrossRef]
- 2. Baker, D.A.; Algorta, G.P. The relationship between online social networking and depression: A systematic review of quantitative studies. *Cyberpsychol. Behav. Soc. Netw.* **2016**, *19*, 638–648. [CrossRef] [PubMed]
- Hamilton, K.; Thomson, C.E.; White, K.M. Promoting active lifestyles in young children: Investigating mothers' decisions about their child's physical activity and screen time behaviours. *Matern. Child Health J.* 2013, 17, 968–976. [CrossRef] [PubMed]
- 4. Valentine, C. Health Implications of Virtual Architecture: An Interdisciplinary Exploration of the Transferability of Findings from Neuroarchitecture. *Int. J. Environ. Res. Public Health* **2023**, *20*, 2735. [CrossRef] [PubMed]
- Habak, S.; Bennett, J.; Davies, A.; Davies, M.; Christensen, H.; Boydell, K. Edge of the Present: A Virtual Reality Tool to Cultivate Future Thinking, Positive Mood and Wellbeing. *Int. J. Environ. Res. Public Health* 2020, 18, 140. [CrossRef] [PubMed]
- 6. Ulrich, R.S. View through a window may influence recovery from surgery. *Science* **1984**, 224, 420–421. [CrossRef] [PubMed]
- Ulrich, R.S.; Simons, R.F.; Losito, B.; Fiorito, E.; Miles, M.A.; Zelson, M. Stress Recovery during Exposure to Natural and Urban Environments. J. Environ. Psychol. 1991, 11, 201–230. [CrossRef]
- 8. Kellert, S.R.; Heerwagen, J.; Mador, M. (Eds.) *Biophilic Design: The Theory, Science and Practice of Bringing Buildings to Life;* John Wiley: New York, NY, USA, 2008.
- Salingaros, N.A. Biophilia and Healing Environments. Healthy Principles for Designing the Built World; Terrapin Bright Green, LLC: New York, NY, USA, 2015; Available online: https://www.terrapinbright-green.com/report/biophilia-healing-environments/ (accessed on 16 August 2023).
- 10. Salingaros, N.A. The Biophilic Index Predicts Healing Effects of the Built Environment. J. Biourbanism 2019, 8, 13–34.
- 11. Salingaros, N. Happiness and Biophilic Urban Geometry. J. Biourbanism 2021, 9, 23–28.
- 12. Gaekwad, J.; Sal-Moslehian, A.; Roös, P.B.; Walker, A. A Meta-Analysis of Emotional Evidence for the Biophilia Hypothesis and Implications for Biophilic Design. *Front. Psychol.* **2022**, *13*, 750245. [CrossRef]
- 13. Mehaffy, M.W.; Salingaros, N.A. A Vision for Architecture as More than the Sum of Its Parts. 2013. Available online: https://www.resilience.org/stories/2013-11-13/a-vision-for-architecture-as-more-than-the-sum-of-its-parts/ (accessed on 16 August 2023).
- 14. Hajrasoulih, A.; del Rio, V.; Francis, J.; Edmondson, J. Urban form and mental wellbeing: Scoping a theoretical framework for action. *J. Urban Des. Ment. Health* **2018**, *5*, 10.
- 15. Cuppers, K. Why We Like to Blame Buildings. 2015. Available online: https://blog.oup.com/2015/08/france-postwararchitecture/ (accessed on 16 August 2023).
- 16. Moutselos, M. Fighting for Their Neighborhood: Urban Policy and Anti-State Riots in France. *Soc. Forces* **2020**, *98*, 1719–1743. [CrossRef]
- 17. Mariotti, A. The effects of chronic stress on health: New insights into the molecular mechanisms of brain-body communication. *Future Sci. OA* 2015, *1*, FSO23. [CrossRef] [PubMed]
- 18. Alexander, C. The Nature of Order. Book 2: The Process of Creating Life; Center for Environmental Structure: Berkeley, CA, USA, 2002.
- Salingaros, N.A.; Mehaffy, M.W. Geometrical Fundamentalism, Chapter 9. In A Theory of Architecture, 2nd ed.; Sustasis Press: Portland, OR, USA, 2014; pp. 172–194. Available online: https://www.academia.edu/5074196/Geometrical_fundamentalism (accessed on 9 July 2022).
- Barton, J.; Pretty, J. What is the Best Dose of Nature and Green Exercise for Improving Mental Health? A Multi-Study Analysis. Environ. Sci. Technol. 2010, 44, 3947–3955. [CrossRef] [PubMed]
- 21. Ewert, A.; Chang, Y. Levels of Nature and Stress Response. Behav. Sci. 2018, 8, 49. [CrossRef]
- 22. Kaplan, S. The restorative benefits of nature: Toward an integrative framework. J. Environ. Psychol. 1995, 15, 169–182. [CrossRef]

- 23. Berman, M.G.; Jonides, J.; Kaplan, S. The Cognitive Benefits of Interacting with Nature. *Psychol. Sci.* 2008, 19, 1207–1212. [CrossRef]
- 24. Berman, M.G.; Kross, E.; Krpan, K.; Askren, M.; Burson, A.; Deldin, P.; Kaplan, S.; Sherdell, L.; Gotlib, I.; Jonides, J. Interacting with nature improves cognition and affect for individuals with depression. *J. Affect. Disord.* **2012**, *140*, 300–305. [CrossRef]
- 25. Berto, R. Exposure to restorative environments helps restore attentional capacity. J. Environ. Psychol. 2005, 25, 249–259. [CrossRef]
- 26. Kardan, O.; Gozdyra, P.; Misic, B.; Moola, F.; Palmer, L.J.; Paus, T.; Berman, M.G. Neighborhood greenspace and health in a large urban center. *Sci. Rep.* **2015**, *5*, 11610. [CrossRef]
- 27. Pretty, J.; Peacock, J.; Hine, R.; Sellens, M.; South, N.; Griffin, M. Green exercise in the UK countryside: Effects on health and psychological well-being, and implications for policy and planning. *J. Environ. Plan. Manag.* 2007, *50*, 211–231. [CrossRef]
- 28. Mehaffy, M.; Salingaros, N. Biophilia: Our Evolved Kinship to Biological Forms. In *Unified Architectural Theory*; Sustasis Press: Portland, OR, USA, 2013; Chapter 17; Available online: https://patterns.architexturez.net/doc/az-cf-172727 (accessed on 16 August 2023).
- Kuo, F.E.; Sullivan, W.C. Environment and Crime in the Inner City: Does Vegetation Reduce Crime? *Environ. Behav.* 2001, 33, 343–367. [CrossRef]
- Kuo, F.E.; Sullivan, W.C. Aggression and Violence in the Inner City: Effects of Environment via Mental Fatigue. *Environ. Behav.* 2001, 33, 543–571. [CrossRef]
- 31. Frumkin, H. Beyond toxicity: Human health and the natural environment. AJPM—Am. J. Prev. Med. 2001, 20, 234–240. [CrossRef]
- 32. Goldberger, A.L. Fractals and the birth of Gothic: Reflections on the biologic basis of creativity. *Mol. Psychiatry* **1996**, *1*, 99–104. [PubMed]
- 33. Taylor, R.P. Reduction of Physiological Stress Using Fractal Art and Architecture. Leonardo 2006, 39, 245–251. [CrossRef]
- 34. Salingaros, N.A. Fractal Art and Architecture Reduce Physiological Stress. In *Unified Architectural Theory*; Sustasis Press: Portland, OR, USA, 2013; Chapter 26.
- 35. Alexander, C. The Phenomenon of Life: The Nature of Order, Book 1; Center for Environmental Structure: Berkeley, CA, USA, 2001.
- 36. Salingaros, N. Unified Architectural Theory: Form, Language, Complexity: A Companion to Christopher Alexander's "the Phenomenon of Life: The Nature of Order, Book 1"; Sustasis Foundation: Portland, OR, USA, 2013.
- Yeo, N.; White, M.; Alcock, I.; Garside, R.; Dean, S.; Smalley, A.; Gatersleben, B. What is the best way of delivering virtual nature for improving mood? *J. Environ. Psychol.* 2020, 72, 101500. [CrossRef] [PubMed]
- Browning, M.; Mimnaugh, K.J.; van Riper, C.J.; Laurent, H.K.; LaValle, S. Can Simulated Nature Support Mental Health? Comparing Short, Single-Doses of 360-Degree Nature Videos in Virtual Reality with the Outdoors. *Front. Psychol.* 2020, 10, 2667. [CrossRef] [PubMed]
- Browning, M.H.E.M.; Shin, S.; Drong, G.; McAnirlin, O.; Gagnon, R.J.; Ranganathan, S.; Sindelar, K.; Hoptman, D.; Bratman, G.N.; Yuan, S.; et al. Daily exposure to virtual nature reduces symptoms of anxiety in college students. *Sci. Rep.* 2023, 13, 1239. [CrossRef]
- 40. Frigione, I.; Massetti, G.; Girondini, M.; Etzi, R.; Scurati, G.; Ferrise, F.; Chirico, A.; Gaggioli, A.; Gallace, A. An Exploratory Study on the Effect of Virtual Environments on Cognitive Performances and Psychophysiological Responses. *Cyberpsychol. Behav. Soc. Netw.* **2022**, *25*, 666–671. [CrossRef]
- Li, H.; Du, X.; Ma, H.; Wang, Z.; Li, Y.; Wu, J. The Effect of Virtual-Reality-Based Restorative Environments on Creativity. *Int. J. Environ. Res. Public Health* 2022, 19, 12083. [CrossRef] [PubMed]
- 42. Chan, S.H.M.; Qiu, L.; Esposito, G.; Mai, K.P.; Tam, K.-P.; Cui, J. Nature in virtual reality improves mood and reduces stress: Evidence from young adults and senior citizens. *Virtual Real.* **2021**, 1–16. [CrossRef] [PubMed]
- Mollazadeh, M.; Zhu, Y. Application of Virtual Environments for Biophilic Design: A Critical Review. *Buildings* 2021, 11, 148. [CrossRef]
- 44. Bullinger, H.J.; Bauer, W.; Wenzel, G.; Blach, R. Towards user centred design (UCD) in architecture based on immersive virtual environments. *Comput. Ind.* **2010**, *61*, 372–379. [CrossRef]
- 45. Heydarian, A.; Carneiro, J.P.; Gerber, D.; Becerik-Gerber, B.; Hayes, T.; Wood, W. Immersive virtual environments versus physical built environments: A benchmarking study for building design and user-built environment explorations. *Autom. Constr.* **2015**, *54*, 116–126. [CrossRef]
- Shemesh, A.; Leisman, G.; Bar, M.; Grobman, Y.J. A neurocognitive study of the emotional impact of geometrical criteria of architectural space. *Archit. Sci. Rev.* 2021, 64, 394–407. [CrossRef]
- Shemesh, A.; Leisman, G.; Bar, M.; Grobman, Y.J. The emotional influence of different geometries in virtual spaces: A neurocognitive examination. J. Environ. Psychol. 2022, 81, 101802. [CrossRef]
- 48. Lee, Y.; Chen, A.N.K. Usability design and psychological ownership of a virtual world. *J. Manag. Inf. Syst.* **2011**, *28*, 269–308. [CrossRef]
- Morris, C. Nintendo Sells 10 Million Copies of the Legend of Zelda: Tears of the Kingdom in 3 Days. Fortune. 17 May 2023. Available online: https://fortune.com/2023/05/17/nintendo-legend-of-zelda-tears-of-the-kingdom-sales/ (accessed on 26 June 2023).
- The Legend of Zelda: Tears of the Kingdom. 12 May 2023. Available online: https://www.metacritic.com/game/switch/the-legend-of-zelda-tears-of-the-kingdom (accessed on 26 June 2023).
- Petit, C. The Legend of Zelda: Tears of The Kingdom. 23 June 2023. Available online: https://kotaku.com/the-legend-of-zeldatears-of-the-kingdom-the-kotaku-r-1850570840 (accessed on 26 June 2023).

- 52. Peters, J. How Many Hours Have You Played Tears of the Kingdom? 22 May 2023. Available online: https://www.theverge.com/ 23732841/legend-of-zelda-tears-of-the-kingdom-hours-played-check# (accessed on 26 June 2023).
- Piazza, J. Audiences Experience 'Avatar' Blues. 11 January 2010. Available online: https://www.cnn.com/2010/SHOWBIZ/ Movies/01/11/avatar.movie.blues/index.html (accessed on 26 June 2023).
- Murphy, J.K. The Beauty of 'Avatar' Left Some Fans Depressed—After Forming a Supportive Online Community, They Now Brace for 'The Way of Water.' 14 December 2022. Available online: https://variety.com/2022/film/features/avatar-depressionfans-the-way-of-water-1235460636/ (accessed on 26 June 2023).
- Lee, I.F. Avatar: Pandora's Aesthetics of Joy. 19 January 2010. Available online: https://aestheticsofjoy.com/avatar-pandorasaesthetics-of-joy/ (accessed on 26 June 2023).
- 56. Halsted, D. The Origins of the Architectural Metaphor in Computing: Design and Technology at IBM, 1957–1964. *IEEE Ann. Hist. Comput.* **2018**, 40, 61–70. [CrossRef]
- 57. Mehaffy, M.W.; Salingaros, N.A. The Pattern Technology of Christopher Alexander. 7 October 2011. Available online: https://metropolismag.com/viewpoints/the-pattern-technology-of-christopher-alexander/ (accessed on 16 August 2023).
- 58. Alexander, C.; Ishikawa, S.; Silverstein, M.; Jacobson, M.; Fiksdahl-King, I.; Angel, S. *A Pattern Language*; Oxford University Press: New York, NY, USA, 1977.
- Mehaffy, M.W.; Kryazheva, Y.; Rudd, A.; Salingaros, N.A.; Gren, A.; Mehaffy, L. A New Pattern Language for Growing Regions: Places, Networks, Processes; Sustasis Press: Portland, OR, USA; Centre for the Future of Places KTH Royal Institute of Technology, Stockholm, Sweden and UN-Habitat: New York, NY, USA, 2020.
- 60. Tendon, S. Alexandrian Patterns: A Solution to a Problem in a Context. 9 February 2016. Available online: https://tameflow. com/blog/2016-02-09/design-patterns/ (accessed on 16 August 2023).
- 61. Wania, C.E. Investigating an author's influence using citation analyses: Christopher alexander (1964–2014). *Proc. Assoc. Inf. Sci. Technol.* **2015**, *52*, 1–10. [CrossRef]
- Kalay, Y.E.; Marx, J. Architecture and the Internet: Designing places in cyberspace. First Monday, Special Issue #5: Virtual Architecture at State of Play III, 2005. In *Reinventing the Discourse, Proceedings of the ACADIA 2001, New York, NY, USA, 11–14* October 2001; Jabi, W., Ed.; Buffalo: New York, NY, USA, 2001; pp. 230–240. [CrossRef]
- 63. Brielmann, A.; Buras, N.; Salingaros, N.; Taylor, R. What happens in your brain when you walk down the street? Implications of architectural proportions, biophilia, and fractal geometry for urban science. *Urban Sci.* 2022, *6*, 3. [CrossRef]
- Taylor, C. Is OnlyFans Ruining Your Love Life? Half of Adult Content Creators Admit They Struggle with Dating or End up Splitting from Their Partner Because of Their Job. 25 April 2023. Available online: https://www.dailymail.co.uk/tvshowbiz/article-12013701/Is-OnlyFans-ruining-love-life-Half-adult-content-creators-admit-struggle-dating.html (accessed on 26 June 2023).
- 65. Lenkert, E. What Is a Content Creator and How to Become One. 10 December 2020. Available online: https://www.adobe.com/ express/learn/blog/content-creator (accessed on 26 June 2023).
- Salingaros, N. Why Monotonous Repetition is Unsatisfying. *arXiv* 2011, arXiv:1109.1461. Available online: https://arxiv.org/pdf/1109.1461v1.pdf (accessed on 16 August 2023).
- 67. Wilkins, A.J.; Penacchio, O.; Leonards, U. The Built Environment and Its Patterns: A View from the Vision Sciences. *J. Sustain. Des. Appl. Res. Innov. Eng. Built Environ.* **2018**, *6*, 42–48.
- Penacchio, O.; Wilkins, A.J. Visual discomfort and the spatial distribution of Fourier energy. Vis. Res. 2015, 108, 1–7. [CrossRef] [PubMed]
- 69. Wilkins, A.J. Looking at Buildings Can Actually Give People Headaches. *Conversation* **2018**, *5*, 1–5. Available online: https://www.cnn. com/style/article/why-looking-at-buildings-can-give-people-headaches/index.html (accessed on 30 December 2021).
- 70. Salingaros, N. A Theory of Architecture, 2nd ed.; Sustasis Press: Portland, OR, USA, 2014.
- Gould, J.D.; Lewis, C. Designing for usability: Key principles and what designers think. *Commun. ACM* 1985, 28, 300–311. [CrossRef]
- 72. Norman, D. The Design of Everyday Things: Updated and Expanded Edition; Basic Books: New York, NY, USA, 2013.
- 73. Yablonski, J. Laws of UX: Using Psychology to Design Better Products & Services; O'Reilly Press: Sebastopol, CA, USA, 2020; pp. 65–76. ISBN 149205531X.
- 74. Sagmeister, S.; Walsh, J. Beauty; Phaidon Press: London, UK, 2018.
- 75. Moore, A. Do Design: Why Beauty is Key to Everything; Chronicle Books: San Francisco, CA, USA, 2019.
- 76. Sharp, H.; Rogers, Y.; Preece, J. Interaction Design: Beyond Human-Computer Interaction, 6th ed.; Wiley: Hoboken, NJ, USA, 2023.
- 77. Mehaffy, M.; Salingaros, N.A. The Surprisingly Important Role of Symmetry in Healthy Places. Planetizen 2021. Available online: https://www.planetizen.com/features/112503-surprisingly-important-role-symmetry-healthy-places (accessed on 16 August 2023).
- 78. Salingaros, N. Life and the Geometry of the Environment. Permac. *Res. Inst.* 2010. Available online: https://www.permaculturenews.org/2010/10/14/life-and-the-geometry-of-the-environment/ (accessed on 16 August 2023).
- 79. Lavdas, A.; Salingaros, N.; Sussman, A. Visual Attention Software: A new tool for understanding the subliminal experience of the built environment. *Appl. Sci.* 2021, *11*, 6197. [CrossRef]
- 80. Lavdas, A.; Salingaros, N. Architectural Beauty: Developing a Measurable and Objective Scale. Challenges 2022, 13, 56. [CrossRef]
- Salingaros, N. Beauty and the Nature of Matter: The Legacy of Christopher Alexander. N. Engl. Rev. 2019. Available online: https://www.newenglishreview.org/articles/beauty-and-the-nature-of-matter-the-legacy-of-christopher-alexander/ (accessed on 16 August 2023).

- Salingaros, N. Connecting to the World: Christopher Alexander's Tool for Human-Centered Design. She Ji J. Des. Econ. Innov. 2020, 6, 455–481. [CrossRef]
- Galle, P. Christopher Alexander's Battle for Beauty: Any Prospects of Victory? She Ji J. Des. Econ. Innov. 2020, 6, 380–385. [CrossRef]
- 84. Scruton, R. Beauty: A Very Short Introduction; Oxford University Press: Oxford, UK, 2011.
- 85. Ruggles, D.H. Beauty, Neuroscience, and Architecture: Timeless Patterns and Their Impact on Our Well-Being; Fibonacci Press: Denver, CO, USA, 2017.
- 86. Buras, N.H. *The Art of Classic Planning: Building Beautiful and Enduring Communities;* Harvard University Press: Cambridge, MA, USA, 2020.
- 87. Lavdas, A.; Mehaffy, M.; Salingaros, N. AI, the Beauty of Places, and the Metaverse: Beyond Geometrical Fundamentalism. *Archit. Intell.* **2023**, *2*, *8*. [CrossRef]
- Smith, M. Why Is Meta's Metaverse Such a Misfire? *IEEE Spectr.* 2022. Available online: https://spectrum.ieee.org/metaversemeta-misfire (accessed on 16 August 2023).
- 89. Stephens, J. Is Santa Clara—The Heart of Silicon Valley—Soulless and Banal by Intent? 20 June 2023. Available online: https://commonedge.org/is-santa-clara-the-heart-of-silicon-valley-soulless-and-banal-by-intent/ (accessed on 16 August 2023).
- 90. Appleton, J. The Experience of Landscape, rev. ed.; Wiley: London, UK, 1996.
- 91. Dosen, A.S.; Ostwald, M.J. Evidence for prospect-refuge theory: A meta-analysis of the findings of environmental preference research. *City Territ. Archit.* **2016**, *3*, 4. [CrossRef]
- Apple. Designing for visionOS. 21 June 2023. Available online: https://developer.apple.com/design/human-interfaceguidelines/designing-for-visionos (accessed on 26 June 2023).
- 93. Apple. Materials. 21 June 2023. Available online: https://developer.apple.com/design/human-interface-guidelines/materials# visionOS (accessed on 26 June 2023).
- 94. Apple. Design for Spatial User Interfaces. 21 June 2023. Available online: https://developer.apple.com/videos/play/wwdc202 3/10076/ (accessed on 26 June 2023).
- 95. Petrigna, L.; Musumeci, G. The Metaverse: A New Challenge for the Healthcare System: A Scoping Review. J. Funct. Morphol. Kinesiol. 2022, 7, 63. [CrossRef]
- 96. Koohsari, M.J.; McCormack, G.R.; Nakaya, T.; Yasunaga, A.; Fuller, D.; Nagai, Y.; Oka, K. The Metaverse, the Built Environment, and Public Health: Opportunities and Uncertainties. *J. Med. Internet Res.* **2023**, *25*, 43549. [CrossRef] [PubMed]
- Fernández-Caballero, A.; Martínez-Rodrigo, A.; Pastor, J.M.; Castillo, J.C.; Lozano-Monasor, E.; López, M.T.; Zangróniz, R.; Latorre, J.M.; Fernández-Sotos, A. Smart environment architecture for emotion detection and regulation. *J. Biomed. Inform.* 2016, 64, 55–73. [CrossRef] [PubMed]
- Marín-Morales, J.; Higuera-Trujillo, J.L.; Greco, A.; Guixeres, J.; Llinares, C.; Scilingo, E.P.; Alcañiz, M.; Valenza, G. Affective computing in virtual reality: Emotion recognition from brain and heartbeat dynamics using wearable sensors. *Sci. Rep.* 2018, *8*, 13657. [CrossRef]
- 99. Jain, A.; Maes, P.; Sra, M. Adaptive Virtual Neuroarchitecture. In *Everyday Virtual and Augmented Reality*; Simeone, A., Weyers, B., Bialkova, S., Lindeman, R., Eds.; Springer: Cham, Switzerland, 2023; pp. 227–249. [CrossRef]
- Elver-Boz, T.; Demirkan, H.; Urgen, B.A. Visual perception of the built environment in virtual reality: A systematic characterization of human aesthetic experience in spaces with curved boundaries. *Psychol. Aesthet. Creat. Arts*, 2022; *Epub ahead of printing*. [CrossRef]
- Coburn, A.; Vartanian, O.; Kenett, Y.N.; Nadal, M.; Hartung, F.; Hayn-Leichsenring, G.; Navarrete, G.; González-Mora, J.; Chatterjee, A. Psychological and neural responses to architectural interiors. *Cortex* 2020, 126, 217–241. [CrossRef]
- Lee, K.-T.; Park, C.-H.; Kim, J.-H. Examination of User Emotions and Task Performance in Indoor Space Design Using Mixed-Reality. *Buildings* 2023, 13, 1483. [CrossRef]
- Juliantino, C.; Nathania, M.P.; Hendarti, R.; Darmadi, H.; Suryawinata, B. The development of virtual healing environment in VR platform. *Procedia Comput. Sci.* 2023, 216, 310–318. [CrossRef]
- 104. Wiederhold, B.K.; Gao, K.; Sulea, C.; Wiederhold, M.D. Virtual reality as a distraction technique in chronic pain patients. *Cyberpsychol. Behav. Soc. Netw.* **2014**, *17*, 346–352. [CrossRef]
- 105. Rizzo, S.; Hartholt, R. Bravemind: Virtual Reality Exposure Therapy. USC Inst. Creat. Technol. 2023. Available online: https://ict.usc.edu/research/projects/bravemind-virtual-reality-exposure-therapy/ (accessed on 16 August 2023).
- 106. Yabanci, O. Historic architecture in tourism consumption. Tour. Crit. 2022, 3, 2–15. [CrossRef]
- 107. Grande, T.R. Traditional architecture gives better sense of well-being than contemporary glass and steel buildings. NMBU Nor. Univ. Life Sci. 2020. Available online: https://partner.sciencenorway.no/architecture-nmbu/traditional-architecture-gives-bettersense-of-well-being-than-contemporary-glass-and-steel-buildings/1631355 (accessed on 16 August 2023).
- Scopelliti, M.; Carrus, G.; Bonaiuto, M. Is it Really Nature That Restores People? A Comparison with Historical Sites with High Restorative Potential. *Front. Psychol.* 2019, *9*, 2742. [CrossRef] [PubMed]
- 109. Coladangelo, L.P. Be Our Guest or Welcome Foolish Mortals? Disney's Invitation to Play and the Delusion/Illusion of Hyperreal, Immersive Documents. *Proc. Doc. Acad.* 2022, 9, 6. [CrossRef]
- 110. Gallo, C. Disney's New Documentary Shows How the Most Creative Minds behind Its Theme Parks Tackle Challenges, and It's Remarkable. 2019. Available online: https://www.inc.com/carmine-gallo/disneys-new-documentary-shows-how-mostcreative-minds-behind-its-theme-parks-tackle-challenges-its-remarkable.html (accessed on 16 August 2023).

- 111. Brittain-Catlin, T. Understanding British Postmodernism. 29 March 2017. Available online: https://www.archdaily.com/867972 /understanding-british-postmodernism-not-what-you-thought (accessed on 16 August 2023).
- 112. Krier, L. The Architecture of Community; Island Press: Washington, DC, USA, 2009.
- Rouchell, M. 21st Century Classicism. 4 January 2015. Available online: https://mrouchell.wordpress.com/2015/01/04/21stcentury-classicism/ (accessed on 25 January 2023).
- Shubow, J. Architecture Continues to Implode: More Insiders Admit the Profession Is Failing. 6 January 2015. Available online: https://www.forbes.com/sites/justinshubow/2015/01/06/architecture-continues-to-implode-more-insiders-admitthe-profession-is-failing/?sh=684c97e64378 (accessed on 25 January 2023).
- 115. Leigh, C. Reviving Classicism: Modernist architecture answers to a cloistered elite—it's time for a change. *City J.* 2020. Available online: https://www.city-journal.org/federal-architecture (accessed on 25 January 2023).
- Westfall, C.W. Why Classical Architecture Better Serves the Public Good Than Modernist Atrocities. 10 February 2020. Available online: https://thefederalist.com/2020/02/10/why-classical-architecture-better-serves-the-public-good-than-modernist-atrocities/ (accessed on 25 January 2023).
- 117. Buday, R. The Confused and Impoverished State of Architectural Research. 27 July 2017. Available online: https://commonedge.org/the-confused-and-impoverished-state-of-architectural-research/ (accessed on 16 August 2023).
- 118. Fard, A. Design Validation: How Do You Know if Your Design is Good. 30 November 2022. Available online: https://adamfard. com/blog/design-validation (accessed on 16 August 2023).
- Mateo, C. Architects and Designers Should Take the Lead in Creating Immersive Environments. 3 March 2023. Available online: https://www.archdaily.com/997175/architects-and-designers-should-take-the-lead-in-creating-immersive-environments? (accessed on 16 August 2023).
- 120. Salingaros, N. Anti-Architecture and Deconstruction: The Triumph of Nihilism, 4th ed.; Sustasis Press: Portland, OR, USA, 2014.
- 121. Masden, K.; Salingaros, N. Intellectual [Dis]honesty in Architecture. J. Archit. Urban. 2014, 38, 187–191. [CrossRef]
- Mehaffy, M.; Salingaros, N. An Obsolete Ideology. *Inference* 2020, 5, 1–4. Available online: https://inference-review.com/letter/ an-obsolete-ideology (accessed on 16 August 2023). [CrossRef]
- 123. Stott, R. 150 Weird Words That Only Architects Use. 19 October 2015. Available online: https://www.archdaily.com/775615/150 -weird-words-that-only-architects-use (accessed on 16 August 2023).
- Agbo, M., Jr. Teacher, Don't Teach Them Nonsense: Reforming Architecture's Broken Education. 14 November 2019. Available online: https://patterns.architexturez.net/doc/az-cf-193230 (accessed on 16 August 2023).
- 125. Pernice, R. Urban Housing for the Masses in East Asia: Structuring the Contemporary Cities in Japan, China and South Korea. In Proceedings of the Archtheo16, Istanbul, Turkey, 27–28 October 2016; pp. 361–369, ISBN 978-605-9207-51-5.
- 126. Soo, Z. China keeping 1 hour daily limit on kids' online games. *AP News*. 19 January 2023. Available online: https://apnews. com/article/gaming-business-children-00db669defcc8e0ca1fc2dc54120a0b8 (accessed on 13 July 2023).
- 127. Dillon, D.; Cai, J. Virtual Reality Greenspaces: Does Level of Immersion Affect Directed Attention Restoration in VR Environments? J. Multidiscip. Sci. J. 2022, 5, 334–357. [CrossRef]
- Sun, C. Envisioning 3D Design Systems for the Metaverse. 7 October 2022. Available online: https://uxdesign.cc/envisioning-3d-design-systems-for-the-metaverse-9d5c39f5ef1c (accessed on 16 August 2023).
- Gorichanaz, T. How 'Living Architecture' Could Help the World Avoid a Soul-Deadening Digital Future. 9 August 2022. Available online: https://theconversation.com/how-living-architecture-could-help-the-world-avoid-a-soul-deadening-digital-future-182831 (accessed on 16 August 2023).
- 130. Díaz, A.; Fernández, A. A pattern language for virtual environments. J. Netw. Comput. Appl. 2000, 23, 291–309. [CrossRef]
- Liszio, S.; Masuch, M. Lost in Open Worlds: Design Patterns for Player Navigation in Virtual Reality Games. In Proceedings of the 13th International Conference on Advances in Computer Entertainment Technology, ACE2016, Osaka, Japan, 9–12 November 2016; pp. 1–7. [CrossRef]
- 132. Alexander, C. The Timeless Way of Building; Oxford University Press: New York, NY, USA, 1979.
- 133. Salingaros, N.A. Why Christopher Alexander failed to humanize architecture. *Side View J.* 2021. Available online: https://thesideview.co/journal/why-christopher-alexander-failed-to-humanize-architecture/ (accessed on 26 July 2023).
- 134. Rennix, A.; Robinson, N.J. Why You Hate Contemporary Architecture. *Curr. Aff.* 2017. Available online: https://www.currentaffairs.org/2017/10/why-you-hate-contemporary-architecture (accessed on 26 July 2023).
- Nuñez, M.; Anderson, D. QWAN: Quality without A Name. The Rabbit Hole: The Definitive Developer's Podcast, Podcast 207, 25 May 2021. Available online: https://therabbithole.libsyn.com/207-qwan-quality-without-a-name (accessed on 16 August 2023).
- 136. Gorichanaz, T. Being at home in the metaverse? Prospectus for a social imaginary. AI Ethics 2023, 3, 647–658. [CrossRef]
- 137. Heidegger, M. Building, dwelling, thinking. In *Poetry, Language, Thought;* Hofstadter, A., Ed.; Harper Perennial: New York, NY, USA, 1971; pp. 143–159.
- 138. Bachelard, G. The Poetics of Space; Beacon Press: Boston, MA, USA, 1994.
- Mehaffy, M. The Impacts of Symmetry in Architecture and Urbanism: Toward a New Research Agenda. *Buildings* 2020, 10, 249.
 [CrossRef]
- 140. Mehaffy, M.; Salingaros, N.A. Symmetry in architecture: Toward an overdue reassessment. *Symmetry Cult. Sci.* **2021**, *32*, 311–343. [CrossRef]
- 141. Salingaros, N.A. Symmetry gives meaning to architecture. Symmetry Cult. Sci. 2020, 31, 231–260. [CrossRef]

- 142. Martins, M.J.; Fischmeister, F.P.; Puig-Waldmüller, E.; Oh, J.; Geissler, A.; Robinson, S.; Fitch, W.T.; Beisteiner, R. Fractal image perception provides novel insights into hierarchical cognition. *Neuroimage* **2014**, *96*, 300–308. [CrossRef]
- 143. Fischmeister, F.P.; Martins, M.J.D.; Beisteiner, R.; Fitch, W.T. Self-similarity and Recursion as Default Modes in Human Cognition. *Cortex* 2016, *97*, 183–201. [CrossRef]
- 144. Le, A.T.D.; Payne, J.; Clarke, C.; Kelly, M.A.; Prudenziati, F.; Armsby, E.; Penacchio, O.; Wilkins, A.J. Discomfort from urban scenes: Metabolic consequences. *Landsc. Urban Plan.* **2017**, *160*, 61–68. [CrossRef]
- 145. Wilkins, A.J. A physiological basis for visual discomfort: Application in lighting design. *Light. Res. Technol.* **2016**, *48*, 44–54. [CrossRef]
- Williams, D. Groups and Goblins: The Social and Civic Impact of an Online Game. J. Broadcast. Electron. Media 2006, 50, 651–670.
 [CrossRef]
- 147. Zeki, S. Beauty in Architecture: Not a Luxury—Only a Necessity. Special Issue: Beauty Matters: Human Judgement and the Pursuit of New Beauties in Post-Digital Architecture. *Archit. Des.* **2019**, *89*, 14–19. [CrossRef]
- 148. Yang, T.; Formuli, A.; Paolini, M.; Zeki, S. The neural determinants of beauty. Eur. J. Neurosci. 2022, 1, 91–106. [CrossRef]
- 149. Ishizu, T.; Zeki, S. Toward A Brain-Based Theory of Beauty. PLoS ONE 2011, 6, e21852. [CrossRef]
- Valentine, C. Architectural Allostatic Overloading: Exploring a Connection between Architectural Form and Allostatic Overloading. Int. J. Environ. Res. Public Health 2023, 20, 5637. [CrossRef]
- 151. McCrae, R.R.; John, O.P. An Introduction to the Five-Factor Model and Its Applications. *J. Personal.* **1992**, *2*, 175–215. [CrossRef] [PubMed]
- 152. Costa, P.; McCrae, R.R. Neo PI-R Professional Manual; Psychological Assessment Resources: Odessa, FL, USA, 1992; p. 396.
- 153. Costa, P.; McCrae, R.R. The Five-Factor Model and the NEO Inventories. In *Oxford Handbook of Personality Assessment*; Butcher, J.N., Ed.; Oxford University Press: Oxford, UK, 2012; pp. 299–322. [CrossRef]
- 154. Haidt, J.; Keltner, D. Appreciation of beauty and excellence. In *Character Strengths and Virtues. A Handbook of Classification;* Peterson, C., Seligman, M.E.P., Eds.; American Psychological Association: Washington, DC, USA, 2004; pp. 537–552.
- 155. Diessner, R.; Solom, R.D.; Frost, N.K.; Parsons, L.; Davidson, J. Engagement with Beauty: Appreciating Natural, Artistic, and Moral Beauty. J. Psychol. Interdiscip. Appl. 2008, 142, 303–332. [CrossRef] [PubMed]
- 156. Lee, E.-J.; Lee, W.; Bae, I. What Is the Draw of the Metaverse? Personality Correlates of Zepeto Use Motives and Their Associations with Psychological Well-Being. *Cyberpsychol. Behav. Soc. Netw.* **2023**, *26*, 161–168. [CrossRef] [PubMed]
- 157. Joshi, A.; Das, S.; Sekar, S. How Big Five Personality Traits affect Information and Communication Technology Use: A Meta-Analysis. *Australas. J. Inf. Syst.* 2023, 27, 1–38. [CrossRef]
- 158. Jojo, C.E.; Sundaramoorthy, J. Personality traits associated with Internet addiction among college students in South India. *Cogent Educ.* 2022, *9*, 2142455. [CrossRef]
- Yang, B.; Wei, L.; Pu, Z. Measuring and Improving User Experience through Artificial Intelligence-Aided Design. *Front. Psychol.* 2020, 11, 595374. [CrossRef] [PubMed]
- 160. Barker, N. Altopia, Dezeen. 2023. Available online: https://www.dezeen.com/aitopia/ (accessed on 16 August 2023).
- Salingaros, N. ChatGPT Identifies Inhuman Architecture—But Can We Please Stop Building It? N. Engl. Rev. 2023. Available online: https://www.newenglishreview.org/articles/chatgpt-identifies-inhuman-architecture-but-can-we-please-stop-buildingit/ (accessed on 16 August 2023).
- 162. Salingaros, N. Even ChatGPT Knows Classical Architecture Is Better than Modern Monstrosities for Federal Buildings. 2023. Available online: https://thefederalist.com/2023/07/06/even-chatgpt-knows-classical-architecture-is-better-than-modern-monstrosities-for-federal-buildings/ (accessed on 16 August 2023).
- 163. Salingaros, N. Architecture's Abysmal Ignorance. Artificial Intelligence Reveals What Experts Deny. 2022. Available online: https://thecritic.co.uk/architectures-abysmal-ignorance/ (accessed on 16 August 2023).
- 164. Curl, J.S. Making Dystopia: The Strange Rise and Survival of Architectural Barbarism; Oxford University Press: Oxford, UK, 2018.
- Engstrom, D.F. Corralling capture. *Harv. J. Law Public Policy* 2013, 36, 31–39. Available online: https://www.proquest.com/ docview/1470800289 (accessed on 16 August 2023).
- Mehaffy, M.; Salingaros, N.A. Architectural myopia: Designing for industry, not people (The Architect Has No Clothes). *Commons* 2011. Available online: https://www.onthecommons.org/magazine/architect-has-no-clothes/ (accessed on 16 August 2023).
- Chang, E.; Kim, H.T.; Yoo, B. Virtual Reality Sickness: A Review of Causes and Measurements. Int. J. Hum. –Comput. Interact. 2020, 36, 1658–1682. [CrossRef]
- 168. Tian, N.; Lopes, P.; Boulic, R. A review of cybersickness in head-mounted displays: Raising attention to individual susceptibility. *Virtual Real.* **2022**, *26*, 1409–1441. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.